Author's response to reviews

Title: Evaluation of interventions on road traffic injuries in Peru: a qualitative approach

Authors:

Luis Huicho (luicho@gmail.com)
Taghreed Adam (adamt@who.int)
Edmundo Rosales (edmundo_rosales@gmail.com)
Ada Paca (ada.paca@gmail.com)
Luis Lopez (llopez.29.06@gmail.com)
Diego Luna (dluna10@gmail.com)
J. Jaime Miranda (mirandaji@gmail.com)
Pablo Best (pablo.best@upch.pe)
Miriam Egusquiza (miriamegu@gmail.com)
Claudia Lema (claudialema@gmail.com)
Esperanza Ludeña (espelu@gmail.com)

Version: 2 Date: 17 September 2011

Author's response to reviews:

Lima, September 16, 2011

MS: 4575159445928424

Audrey Ann Reyes
Journal Editorial Office
BioMed Central
BMC Public Health

Dear Editor,

I am resubmitting a revised version of the manuscript entitled “Evaluation of interventions on road traffic injuries in Peru: a qualitative approach”, with the modifications requested by the Editorial Office. We have specifically detailed in Methods section the qualitative analysis theoretical framework, analysis process, and specific details of the thematic categories obtained from the interviews. They appear now in page 5 (last paragraph), page 6 (first paragraph), in page 6 (last paragraph), and in page 7 (first, second, and third paragraph).

Yours sincerely,

Luis Huicho